Author's response to reviews

Title: Clinical correlation between premature ovarian failure and chromosomal anomaly in a 22-year-old Caucasian woman: case report

Authors:

Domenico Dell'Edera (ducati98@libero.it)
Andrea Tinelli (aditrani@hotmail.it)
Oronzo Capozzi (aedra@libero.it)
Annunziata Anna Epifania (alfchiacchio@tiscali.it)
Antonio Malvasi (amastroberti@innotec-srl.it)
Dominga Lofrese (angelo.capozzo@fastwebnet.it)
Elena Pacella (annalisamarsano82@live.it)
Manuela Leo (annaselvaggi@libero.it)
Mariano Rocchi (antonio.balzano1975@libero.it)

Version: 2 Date: 12 June 2012

Author's response to reviews: see over
Dear Editor,
on behalf of my co-authors i submit the enclosed manuscript for consideration to the Journal of Medical Case Reports. I confirm that the work has not been published in this or a substantially similar form (in print or electronically, including on a website), nor accepted for publication elsewhere, nor is it under consideration by another publication at this time. This work, named “Clinical correlation between premature ovarian failure and chromosomal anomaly in a 22-year-old Caucasian woman: case report”, has underlined that karyotyping is one of the fundamental investigations in the evaluation of amenorrhea. It has highlighted of the chromosomal abnormality, one of the genetic etiology as the causal factor in amenorrhea (POF: Premature ovarian failure).

Best Regards,
Domenico Dell'Edera

REFERENCES:
Dott. Domenico Dell’Edera
Unit of Cytogenetic and Molecular Genetics
“Madonna delle Grazie” Hospital, Matera (Italy).
MAIL TO: ducati98@libero.it.